Author's response to reviews

**Title:** Feasibility, design and conduct of a pragmatic randomized controlled trial to reduce overweight and obesity in children: The Electronic Games to Aid Motivation to Exercise (eGAME) study

**Authors:**

- Ralph Maddison (r.maddison@ctru.auckland.ac.nz)
- Louise Foley (l.foley@ctru.auckland.ac.nz)
- Cliona Ni Mhurchu (c.nimhurchu@ctru.auckland.ac.nz)
- Andrew Jull (a.jull@ctru.auckland.ac.nz)
- Yannan Jiang (y.jiang@ctru.auckland.ac.nz)
- Harry Prapavessis (hprapave@uwo.ca)
- Anthony Rodgers (arodgers@george.org.au)
- Stephen Vander Hoorn (s.vanderhoorn@ctru.auckland.ac.nz)
- Maea Hohepa (maea.hohepa@sparc.org.nz)
- David Schaaf (d.schaaf@auckland.ac.nz)

**Version:** 3  **Date:** 15 May 2009

**Author's response to reviews:** see over
15 May 2009

To Whom It May Concern:

Please find enclosed a revised manuscript entitled “Feasibility, design and conduct of a pragmatic randomized controlled trial to reduce overweight and obesity in children: The electronic games to aid motivation to exercise (eGAME) study”.

The following changes have been made:

- The reference list has been edited so that it complies with BMC Public Health regulations
- Capitalisation in the title has been removed
- The figure legend has been moved to below the reference list

I look forward to receiving your response.

Regards,

Louise Foley
Research Fellow